Author's response to reviews

Title: An Audit of Influenza and Pneumococcal Vaccination in Rheumatology Outpatients

Authors:

Evin Sowden (evinsowden@hotmail.com)
William Mitchell (william.mitchell@mbht.nhs.uk)

Version: 2 Date: 13 March 2007

Author's response to reviews:

Minor change to the title (addition of 'An' at the beginning of title) and insertion of the sections 'Abstract', 'Competing interests', 'Authors' contributions' and 'Tables and captions' to the main manuscript.